Author’s response to reviews

Title: Intubation of non-difficult airways using video laryngoscope versus direct laryngoscope: a randomized, parallel-group study

Authors:
De-Xing Liu (pokowenl@163.com)
Ying Ye (627140381@qq.com)
Yu-Hang Zhu (zhuyuhang_zmc@sina.com)
Jing Li (249972159@qq.com)
Hong-Ying He (252769633@qq.com)
Liang Dong (donglqqsx2008@163.com)
Zhaoqiong Zhu (ganzhuzhaoqiong@hotmail.com)

Version: 4 Date: 27 Mar 2019

Author’s response to reviews:

BANE-D-18-00545R3

Intubation of non-difficult airways using glidescope versus direct laryngoscope: a randomized, double-blind, parallel-group study

BMC Anesthesiology

Dear Dr Ahmed Abdelaal Ahmed Mahmoud,

We are grateful to the valuable comments raised by the editors and the reviewers. The manuscript has been revised according to these comments. The revisions are marked using the MS Word track changes function.

Best wishes,
Zhao-Qiong Zhu
Response to the comments:

Please, change the term "Glidescope" to "Video laryngoscope" throughout the manuscript including the title, tables and figure 1.

Much appreciated

Response: Thanks for the suggestion. We have made the necessary changes.